Author's response to reviews

Title: Magnetic Resonance Spectroscopy of normal appearing white matter in early relapsing-remitting Multiple Sclerosis: correlations between disability and spectroscopy

Authors:

Dr Juan Luis Ruiz Pena (jlruizpe@hotmail.com)
Pilar Pinero (havenot@hotmail.com)
Guillermo Sellers (havenot@hotmail.com)
Joaquin Argente (havenot@hotmail.com)
Alfredo Casado (havenot@hotmail.com)
Joaquin Foronda (havenot@hotmail.com)
Antonio Ucles (havenot@hotmail.com)
Guillermo Izquierdo (gizquierdo@supercable.es)

Version: 4  Date: 4 Jun 2004

Dear sir,

We have changed all your suggestion with great interest except (References 8 and 29 - Please include the list of all of the authors, rather than writing et al) because in pubmed there is no information about the lasts authors.

Best Regards,

Juan Luis Ruiz Pena